Author’s response to reviews

Title: First characterization of methanogens in oral cavity in Malian patients with oral cavity pathologies.

Authors:

Elisabeth Sogodogo (elisabethsogodogo@yahoo.fr)
Ogobara Doumbo (okd@icermali.org)
Gérard Aboudharam (gerard.aboudharam@univ-amu.fr)
Bourema Kouriba (kouriba@icermali.org)
Ousseynou Diawara (ussynu@gmail.com)
Hapssa Koita (hapssakoita68@gmail.com)
Souleymane Togora (souleymanetogora@yahoo.fr)
Michel Drancourt (michel.drancourt@univ-amu.fr)

Version: 4 Date: 14 Oct 2019

Author’s response to reviews:

Dear Editor,

Enclosed a revised version of our manuscript following your comments:

1. Authorship

Authors’ response: No authorship modification.

Sincerely,

Prof. Michel Drancourt, M.D., Ph.D.
Corresponding author

Editor Comments :

1. Figure 1

Authors’ response: Sorry for this missing, Figure 1 enclosed.
2. Sequencing data

Authors’ response: As previously acknowledged sequences being 100% identical to homologous references in GenBank, have not been submitted nor stored in the laboratory. Only original sequences (i.e. different by at least one nucleotide) are preserved/submitted (Page 8, Lines 194-196).